Author’s response to reviews

Title: "The sun keeps rising but darkness surrounds us": a qualitative exploration of the lived experiences of women with obstetric fistula in Ethiopia

Authors:

shifera yedenekal (shiferaa@yahoo.com)

Misganaw Animut (misganawanimit@gmail.com)

Abebe Gebretsadik (abema24@yahoo.com)

Lakew Gebretsadik (lakewgebretsadika9@gmail.com)

Million Berhe (megaabera@yahoo.com)

Zewdie Koricha (zbkoricha@yahoo.com)

Version: 7 Date: 03 Jan 2019

Author’s response to reviews:

Response to Reviewer

Hello Again, on behalf of all authors; I want to pass my acknowledgment for reviewer.

Here are the reviewer’s comments and authors’ response

Reviewer 1

Solina Richter (Reviewer 1):

Here are the questions /comments by reviewer and response by authors.

The reviewer gives grammar and word corrections on the following section.

1. Abstract, conclusion section: All corrections are considered see the track change on the main document.

2. Background on page 2&3: word correction and adjective insertion are accepted and modified accordingly. See the track change
3. Method on page 4: All suggested modifications are accepted and rewrite accordingly. See the track change

4. Methods on page 5: All suggested modifications are accepted and rewrite accordingly. See the track change

5. Trustworthiness on page 6. All the recommendations are accepted and being modified. See the track change

6. Result on page 8. All the suggestions are considered. See the track change. The authors also convinced to collapse all sub categories as suggested by the reviewer to maintain the flow of the reading. See the track change

7. Implication and declaration section on page 18: all the suggestions are accepted. See the track change

Response to editor

Dear Editor

Thank you for your valuable comments. We revised the document as per your suggestion.